Author's response to reviews

Title: The modified dental anxiety scale: UK general public population norms in 2008 with further psychometrics and effects of age

Authors:

Gerry M Humphris (gmh4@st-and.ac.uk)
Tom A Dyer (T.Dyer@sheffield.ac.uk)
Peter G Robinson (peter.g.robinson@sheffield.ac.uk)

Version: 7 Date: 25 August 2009

Author's response to reviews:

I have proof read the article and found some small changes as follows:

1) PAGE 5 and last but one line I have included a space between UK and [9].
2) PAGE 6 LINE 8, I have deleted the word 'identified' and replaced with 'previously stated'.
3) I have removed the citation to reference [15] on PAGE 6, 2nd LINE of 2nd PARA. This was incorrect. It does not change the rest of the reference numbering.
4) I have included a comma between 0.957 and 95%CI on PAGE 12, LINE 5.
5) I have included the word 'scoring' on PAGE 16 LINE 10 2nd PARA.